Author's response to reviews

Title: Delayed-onset heparin-induced thrombocytopenia presenting with multiple arteriovenous thromboses: Case report

Authors:

Abbas Salehiomran (abbasomran2006@yahoo.com)
Abbassali Karimi (akarimi@sina.tums.AC.IR)
Hossein Ahmadi (dr.ahmadi2006@yahoo.com)
Parin Yazdanifard (parinyazdanifard@yahoo.com)

Version: 7 Date: 23 October 2007

Author's response to reviews:

23-10- 2007

Dear Editor in chief
The journal of medical case report

Thank you for your email. All of the changes have been made as your suggestion and are listed below. Also the changes are found in my revised manuscript as red font.

1- The author's name have been corrected identically and written in order in which they are listed in the submission system.

2- Email of each author has been added beneath the corresponding details.

3- Consent has been put under the acknowledgments.

4- The order of references has been corrected.

Also we correct some details of title pages like address and affiliations.

Sincerely yours
Abbas. SalehiOmran